Author's response to reviews

Title: Clone-specific Expression, Transcriptional Regulation, and Action of Interleukin-6 in Human Colon Carcinoma Cells

Authors:

- Wolfgang Brozek (wolfgang.brozek@meduniwien.ac.at)
- Giovanna Bises (giovanna.bises@meduniwien.ac.at)
- Gerhild Fabjani (gerhild.fabjani@meduniwien.ac.at)
- Heide S. Cross (heide.cross@meduniwien.ac.at)
- Meinrad Peterlik (meinrad.peterlik@meduniwien.ac.at)

Version: 3 Date: 10 December 2007

Author's response to reviews: see over
Re: Ms. #5869405091573053

Dear Dr. Lê:

We were very pleased to learn that our paper on “Clone-specific Expression, Transcriptional Regulation, and Action of Interleukin-6 in Human Colon Carcinoma Cells, by W. Brozek et al. has been accepted in principle for publication in BMC Cancer. We checked the manuscript carefully to ensure that it now conforms to the journal style.

Please, note that all the experiments described in the paper were done with established clonal cell lines, and that therefore it was not necessary to obtain ethical approval for the study.

Please accept our sincere thanks for the speedy and efficient handling of our manuscript. We also want the Editors to know how much we appreciated the open and fair review process.

(Meinrad Peterlik)